Author’s response to reviews

Title: Safety and efficacy of Holmium laser enucleation of the prostate (HoLEP) in patients with previous transperineal biopsy (TPB): outcomes from a dual-centre case-control study

Authors:

Chris Bell (Christopher.bell@wales.nhs.uk;cjsbell@doctors.org.uk)
Sacha Moore (sacha.moore@wales.nhs.uk)
Amarit Gill (amarit@doctors.org.uk)
Obinna Obi-Njoku (obi-njoku.joseph@wales.nhs.uk)
Stephen Hughes (stephen.hughes6@wales.nhs.uk)
Asad Saleemi (asad.saleemi@ldh.nhs.uk)
Gidon Ellis (super_iqi@yahoo.com)
Farooq Khan (mrfarooqkhan.secretary@btinternet.com)
Iqbal Shergill (iqbal.shergill@wales.nhs.uk)

Version: 1 Date: 06 Jun 2019

Author’s response to reviews:

Dear Tillie Cryer,

Corrections implemented as requested. For clarity we have removed reference to clinical effectiveness team - they are the ones we first contacted about this project but are part of the larger approval committee which includes research and development team etc. It is more correct to say the study was approved by the 'Health Board' so this has been updated.

Thanks,

Chris Bell